Author's response to reviews

Title: Multifocal skeletal tuberculosis in an immunocompetent patient: a case report

Authors:

Paulo Sérgio R Araújo (psergio@cpqam.fiocruz.br)
Heloisa Ramos L Melo (helramos@terra.com.br)
Fábio L Melo (fabio@cpqam.fiocruz.br)
Zulma Medeiros (meideiros@cpqam.fiocruz.br)
Maria Amelia Maciel (amaciel57@gmail.com)
Renata Florêncio (reps77@gmail.com)
Eduardo Brandão (brandaoe@cpqam.fiocruz.br)

Version: 5 Date: 12 June 2015

Author's response to reviews: see over
Dear Editor

Thank you for your consideration. Please be advised that all requested revisions were made. We are sending a new version of the manuscript.

Sincerely,

Eduardo Brandão

-----------------------------------------

1) Abstract sections (please use the following: Background, Case Presentation, Conclusions): Included.

2) Please remove the additional Figure 1 image from the main text: Excluded.